Author’s response to reviews

Title: Dental school tracks related to retention of dentists in Thai government services: a crosssectional survey

Authors:

Philaiporn Vivatbuttsiri (vphilaiporn@yahoo.com)
Thanachok Iempook (thanachok_do@hotmail.com)
Sakda Wonghinkong (sakda_tae@yahoo.com)
Sunisa Sopa (sunisa_sopa@hotmail.com)
Palinee Detsomboonrat (palinee_kung@hotmail.com)

Version: 2 Date: 01 Jan 2020

Author’s response to reviews:

Dear Editor-in-Chief

According to reviewer’s comments we have edited some minor grammar issues and proof it from English language editing service in our University with Native speaker.

All content in this manuscript has been read and approved by the authors. All authors are responsible for the content of manuscript

Thank you very much for consideration, and looking forward to hearing from you.

Sincerely yours,

Palinee